Author's response to reviews

Title: Adult Kawasaki disease in a European patient: a case report and review of the literature

Authors:

Theano Kontopoulou (tkontopoulou@yahoo.gr)
Dimitrios - Georgios Kontopoulos (d.kontopoulos13@imperial.ac.uk)
Emmanouel Vaidakis (vaidakismanolis@gmail.com)
George P. Mousoulis (geomousoulis@yahoo.gr)

Version: 4 Date: 14 December 2014

Author's response to reviews: see over
Dear Journal of Medical Case Reports Editorial Team,

Thank you very much for considering our manuscript entitled “Adult Kawasaki disease in a European patient: a case report and review of the literature” for publication in the Journal of Medical Case Reports. The manuscript ID is 9718460721337305.

We have uploaded a revised version of the manuscript, having addressed the comments made by the editor:

“It should be resubmitted after undergoing a major revision where the unique features of the case is illuminated. The review of the literature section is a simple recitation of a textbook description of the infection and not of other similar cases in Kawasaki disease in young adults.”

✓ We have significantly increased the length of the introduction of the manuscript, citing other published cases of adult Kawasaki disease and discussing the patients’ complications, treatment and long-term outcome. To stress the rarity of the disease, we have also added a global map of published cases of adult Kawasaki disease. Finally, we have elaborated in the discussion section on features of this case that set it apart from most other published ones. All changes in the manuscript are highlighted with red colour.

Sincerely,

Theano Kontopoulou

Dimitrios – Georgios Kontopoulos

Emmanouel Vaidakis

George P. Mousoulis